Author's response to reviews

Title: Dynamic cardiac dyssynchrony is strongly associated with 2-year dialysis adequacy in continuous ambulatory peritoneal dialysis patients

Authors:

Ching-Hui Huang (chinghui@cch.org.tw)
Chia-Chu Chang (chiachu@cch.org.tw)
Tzu-Lan Chang (tzulanchang@ntu.edu.tw)
Yu-Jun Chang (yujun@cch.org.tw)

Version: 3 Date: 21 February 2013

Author's response to reviews: see over
Dear Dr. Hayley Henderson,

We are most grateful for the opportunity to publish our work entitled, “Dynamic cardiac dyssynchrony is strongly associated with 2-year dialysis adequacy in continuous ambulatory peritoneal dialysis patients”. We have substantially revised the manuscript and responded to the comments point-by-point. The minor changes have been highlighted with underline in the revised manuscript.

To ensure that all spelling and grammar is correct in our manuscript, we have used a professional language editing service, Edanz, as your nice comment. Otherwise, we also have gone through the manuscript formatting checklist one more time and ensure that our revised manuscript conforms to all of the points.

We hope the responses to the comments and the revisions to the manuscript have fully addressed your and the Reviewers’ concerns.

Thank you again for consideration of this manuscript.

Sincerely Yours

Chia-Chu Chang, MD
Division of Nephrology, Department of Internal Medicine
Changhua Christian Hospital, Changhua, Taiwan
Responses to Reviewer 1

Point-to-point

Reviewer's report:

1. The LV dyssynchrony index should be described in the Abstract.

Response:

We have the LV dyssynchrony index in the Abstract section: Page 2, Sentence 39th-42nd.

2. The authors conclude in the Abstract “Our study confirmed that cardiac function has an impact on dialysis adequacy”. I think that “cardiac function” should be replaced by “cardiac dysfunction”.

Response:

We have done as your comment: Page 3, sentence 52.

Responses to Reviewer 2

Point-to-point

Reviewer's report:

The main text should be revised by native English language speaker.

Response:

Yes, our manuscript have been edited by a native-English speaker with scientific expertise before remission. We would like to attach the receipt from “Edanz Group Global”.
Hello chia-chu chang,

You sent a payment of $272.70 USD to Edanz Group Global.

This charge will appear on your credit card statement as payment to PAYPAL *EDANZGROUP.

Merchant information  Instruction to merchant
Edanz Group Global  None provided

Shipping information  Shipping method
chia-chu chang  Not specified
No 135 Nanson street Changhua city
Changhua County Taiwan

<table>
<thead>
<tr>
<th>Description</th>
<th>Unit price</th>
<th>Qty</th>
<th>Amount</th>
</tr>
</thead>
<tbody>
<tr>
<td>Editing</td>
<td>$272.70 USD</td>
<td>1</td>
<td>$272.70 USD</td>
</tr>
</tbody>
</table>

Item #: G1312-2606-Chang-Rev-PBP

Total: $272.70 USD

From amount: NT$8,213 TWD
To amount: $272.70 USD

Exchange rate: 1 New Taiwan Dollar = 0.0332034 U.S. Dollars

Receipt No: 0996-1368-0228-6116

Please keep this receipt number for future reference. You'll need it if you contact customer service at Edanz Group Global or PayPal.